Author's response to reviews

Title: p21 is decreased in polycystic kidney disease and leads to increased epithelial cell cycle progression: roscovitine augments p21 levels

Authors:

Jin-young Park (jyppark@ucdavis.edu)
William E Schutzer (schutzer@ohsu.edu)
Jessie N Lindsley (lindsley@ohsu.edu)
Susan P Bagby (bagbys@ohsu.edu)
Terry T Oyama (oyamat@ohsu.edu)
Sharon Anderson (anderssh@ohsu.edu)
Robert H Weiss (rhweiss@ucdavis.edu)

Version: 4 Date: 15 August 2007

Author's response to reviews:

none